Author's response to reviews

Title: Cost-effectiveness of strategies to improve the utilization and provision of maternal and newborn health care in low-income and lower-middle-income countries: a systematic review

Authors:

Lindsay Mangham-Jefferies Mrs (lindsay.mangham-jefferies@lshtm.ac.uk)
Catherine Pitt Ms (catherine.pitt@lshtm.ac.uk)
Simon Cousens Prof (simon.cousens@lshtm.ac.uk)
Anne Mills Prof (anne.mills@lshtm.ac.uk)
Joanna Schellenberg Dr (joanna.schellenberg@lshtm.ac.uk)

Version: 2
Date: 25 March 2014

Author's response to reviews: see over
Two changes have been made:

1) Line numbering: the Manuscript now contains line numbers

2) Ethics: a section has been added towards the end of the document (page 21) to explain ethical approval was not sought since this is a literature review and all the information included in this review has been published and is in the public domain.